لا جمه ما

TarjomeFa.Com

ارائه شده توسط:

سایت ترجمه فا

مرجع جدیدترین مقالات ترجمه شده از نشریات معتبر

The Quarterly Review of Economics and Finance xxx (2013) xxx-xxx



Contents lists available at SciVerse ScienceDirect

The Quarterly Review of Economics and Finance

journal homepage: www.elsevier.com/locate/qref



Information asymmetry and monitoring in equity private placements

Hsiao-Chen Liang^{a,*}, Woan-Yuh Jang^b

- a Department of Business Administration, National Taiwan University of Science and Technology, No. 43, Sec. 4, Keelung Road, Taipei 106, Taiwan, ROC
- ^b Graduate Institute of Finance, National Taiwan University of Science and Technology, No. 43, Sec. 4, Keelung Road, Taipei 106, Taiwan, ROC

ARTICLE INFO

Article history:
Received 16 September 2011
Received in revised form 9 March 2013
Accepted 27 May 2013
Available online xxx

Keywords:
Private placements
Discounts
Announcement effect
Information asymmetry
Ownership structure

ABSTRACT

This paper systematically examines the factors that determine price discounts and announcement effects of equity private placements conducted by firms in Taiwan from 2002 to 2008. Different with most studies of private placements using available observations as a whole sample, our study separates the whole sample into subsamples by exchange-listed firms and OTC firms. The results for OTC firms corroborate the information hypothesis; the discounts serve as compensation for investor's costs of assessing firms, while abnormal returns reflect the information about firm quality. On the other hand, the empirical results show that some of our findings support an information explanation and some support a monitoring explanation in the case of exchange-listed firms. It seems that there are different motives behind the exchange-listed firms placing equity privately.

© 2013 The Board of Trustees of the University of Illinois. Published by Elsevier B.V. All rights reserved.

1. Introduction

The private placement market has escalated rapidly worldwide during the past few years. For example, the amount of private placements in the US (the United States) dramatically increased from \$1.87 billion in 1995 to \$51.9 billion in 2009. Taiwan's private placement market has also expanded annually after the enactment of relevant regulations in 2002. In 2009, issuing firms conducted 181 private placements, amounting to \$5.46 billion, which accounts for 25.36% of the total amount of private and public placements in Taiwan. These figures indicate that private placement has become an important financing alternative for companies seeking capital.

"Private placement" implies offering equity to a selective number of institutions or high wealth investors. Firms choose private placements primarily because the simplified issuance procedures and lower flotation costs expedite issuance (Anderson, Rose, & Cahan, 2006; Fenn, 2000; Krishnamurthy, Spindt, Subramaniam, & Woidtke, 2005). However, this measure could dilute the ownership of current non-participating shareholders. To safeguard these

shareholders' rights, stock exchanges typically enforce specific regulations for private placements (Anderson et al., 2006), including qualification restrictions on investors and resale restrictions on private equity shares. Screening investors may help resolve information asymmetry problems, which are more severe in private placement firms that are new and small, covered by fewer stock market analysts, tending to be listed on the OTC, or deriving most of their value from growth opportunities (Lee & Wu, 2009; Marciukaityte, Szewczyk, & Varma, 2005; Wu, 2004). The private placement resale restrictions affect the pricing of shares. When the majority of block shares are traded at a premium (Barclay & Holderness, 1989; Barclay, Holderness, & Sheehan, 2007; Wayne & Hailu, 1991), private equity has always been offered at a discount. The discounts compensate investors for higher trading costs such as illiquidity (Silber, 1991); thus, investors are able to freely transfer private equity shares only after a holding period of two or three years (lockup period) in most countries. Because investors can't arbitrage by quick turnaround within the lockup period, they must bear the economic risk and thus have the incentives to play the gatekeeping role to monitor firms (Coffee, 1995). This involves the ownership structure hypothesis which suggests that outside blockholders created by private placements offer monitoring services and then increase firm value.

In general, the theories commonly applied to the announcement effects and pricing in private placements include the information hypothesis and the ownership structure hypothesis. However, empirical evidence from different countries or different stock exchanges supports different theories. This makes us wonder

 $1062-9769/\$-see \ front \ matter \ @\ 2013\ The\ Board\ of\ Trustees\ of\ the\ University\ of\ Illinois.\ Published\ by\ Elsevier\ B.V.\ All\ rights\ reserved.\ http://dx.doi.org/10.1016/j.qref.2013.05.009$

Please cite this article in press as: Liang, H. -C., & Jang, W.-Y. Information asymmetry and monitoring in equity private placements. *The Quarterly Review of Economics and Finance* (2013), http://dx.doi.org/10.1016/j.qref.2013.05.009

^{*} Corresponding author. Tel.: +886 2 2737 6753. E-mail addresses: D9408204@mail.ntust.edu.tw (H.-C. Liang), jangwy@mail.ntust.edu.tw (W.-Y. Jang).

¹ Data source: http://www.sagientresearch.com (USA), Securities and Future Bureau, Financial Supervisory Commission (Taiwan). Private placements in USA include PIPEs (private investments in public equity), Regulation S transactions and 144-A placements.

whether the different sample will lead to different results, and which factors influence the results. To explore this, the view of Hertzel and Smith (1993) sheds some light. They suggest that the relative importance between the information hypothesis and the ownership structure hypothesis for private placements depends on firm size. In general, smaller firms tend to be growing, illiquid and with higher managerial ownership, so resolving information asymmetry problems seems more important. In contrast, larger firms are more liquid but with lower managerial ownership, therefore increased monitoring or incentive alignment becomes relatively important.

In Taiwan, there are two main stock markets: centralized market (Taiwan Stock Exchange, TSE) and over-the-counter market (GreTai Securities Market, OTC). Due to the level of listing requirements for securities traded on TSE and OTC being very different, except for the difference in firm size, the governing regulations and the market characteristics between TSE and OTC are also quite different.² Therefore, we consider it valuable to explore the differences between TSE and OTC markets, and the empirical results will be more useful for managers' financing decisions in firms listed on different exchanges. While reviewing the literature, though there is ample empirical evidence to support different hypotheses on private placements, no serious attempt has yet been made to examine whether they apply to both centralized markets and OTC markets. The OTC market exhibits a higher level of information asymmetry than that of the centralized market. As private placements may help mitigate information asymmetry, it would be interesting to explore whether this financing tool is more effective for OTC markets or for centralized markets. In literature, we discovered that only Alli and Thompson (1993) cover exchange-listed and OTC firms but they do not conduct a detailed cross-sectional analysis. Moreover, their results are limited by the low number of sample firms and research variables.

Moreover, differences in regulations among countries also modify empirical results in private placements. Except for resale restrictions, private placement regulations on private equity investors, amount, or discounts are stricter in Singapore than in Taiwan and the US, thus altering the empirical results. For example, there are no limits on private placement amount and discounts in Taiwan and in the US. Empirical results reveal that the maximum equity fraction placed, average discounts, and private placement abnormal returns in these two countries are higher than in Singapore (Chen, Ho, Lee, & Yeo, 2002a; Hertzel & Smith, 1993; Jang, Tsai, & Kuo, 2010; Wruck, 1989). This study divides the entire sample into two subsamples - exchange-listed and OTC firms, and examines the factors that determine private placement discounts and the manner in which investors respond to private placement announcements. Hence, we hope this research will contribute to the literature on private placements in the case of Taiwan.

In conclusion, our results corroborate the information hypothesis in OTC firms. The discounts serve as compensation for the investor's costs of assessing the firm, while abnormal returns reflect the information about firm quality. On the other hand, the empirical results show that some of our findings support an information explanation and some support a monitoring explanation in the

case of the TSE firms. The discounts compensate investors for both information costs and expected monitoring services, while announcement effects reflect both the information about firm quality and the benefits of increased monitoring. Intending to pursue control benefits, single investors participate in private placements in order to obtain control premium. The evidence substantiating this claim is fairly conclusive in exchange-listed firms.

In addition to introduction, the paper is organized as follows: Section 2 presents the regulatory environment and market characteristics in Taiwan. Section 3 reviews the relevant literature on private placements. Section 4 details research methodology. Section 5 contains the sample descriptions and empirical results, and conclusions are presented in Section 6.

2. Regulation environment and market characteristics in Taiwan

2.1. The regulatory environment for private placements

Regulations for private placements in Taiwan primarily follow the pattern of private placements in the US. In contrast to the non-exclusive regulations in the US, regulations for private placements in Taiwan are exclusive. Private placement shares are offered at discounts in most countries; however, they are often sold at a premium in Singapore (Tan, Chng, & Tong, 2002). We thus begin by comparing regulations for private placements in these countries. The primary regulations governing private placements are the Securities and Exchange Act (from Article 43-6 to Article 43-8) in Taiwan, the Regulation D and Rule 144A in the US,³ and the Listing Manual of the Singapore Exchange.

In Taiwan, a public company may conduct a private placement with "accredited investors" only upon shareholder approval. After the date of the shareholders' resolution, the private placement may be conducted by instalments within one year. Accredited investors include financial institutions (banks, bills finance corporation, mutual funds and insurance companies, etc.), wealthy investors (natural persons, juristic persons, or funds meeting the specified wealth standards), and insiders (directors, supervisors, and managerial officers of the company or its affiliated enterprises). There are no limitations on the number of financial institutions, but the number of other investors is limited to 35. In the US, a private placement of securities can be sold to an unlimited number of "accredited investors" and up to 35 "sophisticated investors". However, the restrictions in Singapore are stricter and firms cannot sell shares to their directors and existing blockholders (Anderson et al., 2006; Chen et al., 2002a).

As for the pricing of private placements, the US has no restriction on the size of discount. This was the case in Taiwan before 2005. However, starting from 2005, if the price of privately placed shares was lower than 80% of the reference price,⁴ the offering company had to seek and disclose an independent expert's opinion on the reasonableness of the amount of discount. After 2010, the price of private placement securities was further restricted to 80% of the reference price if insiders were to participate in private placements. In Singapore, the size of the discount is restricted to 10% of the current market price. In addition, in Taiwan and the US,

² In Taiwan, both TSE and OTC use electronic screen trading. If a public company intends to have its shares listed, it may apply for listing either on TSE or on OTC. Compared to TSE, OTC was established to enable firms not meeting the criteria for listing on TSE to access direct financing. Therefore, the requirements for listing on TSE and OTC are different. For example, minimum paid-in capital (NT\$50 million on OTC/NT\$600 million on TSE), no less than two full fiscal years after having completed company registration (OTC/three years on TSE) and requirements for profitability and dispersion of shareholdings. (NT\$ denotes New Taiwan Dollar. US\$1 is around NT\$30 in November, 2012).

³ Private placements to investors outside the US are regulated by the Regulation

⁴ The reference price is the higher of the following two calculations: (a) the firm's average stock price for either the one, three, or five business days before the price determination date, or (b) the firm's average stock price for the thirty business days before the price determination date. (Before 2010, the reference price was only restricted to condition (a).)

there are no restrictions on the amount of a private placement. In Singapore, except in specific circumstances, the volume of shares issued through private placement is restricted to 10% of the shares outstanding (Anderson et al., 2006; Chen et al., 2002a).

In Taiwan, after the sale of privately placed shares, the offering firms must disclose the relevant information as required under the Market Observation Post System (MOPS) within 15 days. This disclosure includes information on the number of shares offered, the offering price, the purpose and investors of the offering. In the US, there are similar regulations for each new offering; the issuer must file a notice of sale with the Securities and Exchange Commission containing the information required by Form D not later than 15 calendar days from the first sale of the securities. In Singapore, the regulation requires immediate disclosure by an issuer to keep the shareholders and the Singapore Exchange informed of the private sale of a significant amount of additional securities.

In addition, there are resale restrictions in Taiwan and the US. but there are no such restrictions in Singapore. In Taiwan, the investors cannot resell privately placed securities within three years as a matter of principle. If the investors intend to transfer the securities within three years, there are restrictions on the buyers and the number of shares that can be transferred. After the three-year holding period, if the company intends to register the privately placed shares, it has to file with the competent authority for retroactive handling of public issuance procedures. In the US, there are two types of private placements, registered (unrestricted) and unregistered (restricted) (Wu, 2004). Investors cannot sell the restricted privately placed securities for at least six months or one year, depending on whether the investors are affiliates or non-affiliates of the issuer (Securities Act Rule 144). However, the securities can be sold to "qualified institutional buyers" or in specified conditions (e.g., notice requirements) without adhering to the holding period (Securities Act Rule 144A). In the latter case, the buyer could be acting in place of an underwriter, with the intention of re-selling the placement.

In summary, private placement of shares in Taiwan can be sold only to accredited investors, the size of the discount is restricted, and the lockup period is longer. Thus, the development of private placements in Taiwan is more restricted than in the US.

2.2. Market characteristics in Taiwan

Several market characteristics unique to Taiwan have made research on its private placements very important. First, compared with firms in developed countries, most firms in Taiwan are rather small. According to data from the Small and Medium Enterprise Administration of Ministry of Economic Affairs, there were 1.23 million small and medium-sized enterprises in Taiwan in 2009, accounting for 97.91% of all business enterprises. Further, since small businesses often consider the higher fixed issuing costs as a barrier to financing and information asymmetry in small firms is also more severe, private placements have become vital for small firms to raise capital (Carpentier, L'Her, & Suret, 2008).

Second, the ratio of family controlled businesses in Taiwan is rather large. According to Lin and Chang (2009), from 1998 to 2006, 57% of the listed firms in Taiwan are family-controlled businesses. Cronqvist and Nilsson (2005) find that family controlled firms avoid conducting private placements because introducing new blockholders may threaten the family's control. However, on the topic of corporate control, Sharpe and Woo (2005) suggest that founders with a low shareholding level will prefer placing their private placement shares with shareholders who share a common interest. Therefore, we conjecture that if family-controlled firms aim to raise funds but are not willing to lose their control rights, the most effective method will be to issue private placement shares to

friendly investors, e.g., the family members. Lin and Chang (2009) also find that family firms in Taiwan are less prone to fraud because the interests of the controlling families and family firms are aligned. In general, the management teams of family firms are often formed by family members⁵ (Yeh, Lee, & Woidtke, 2001). Further, based on the incentive alignment effect, the controlling families also play the gatekeeper role in monitoring firms and thus help reduce the likelihood of fraud.

Third, in contrast to the US market, which mainly consists of institutional investors, most participants in the Taiwanese stock markets are individual investors (66% of all investors in 2011). Institutional investors form expert teams to analyze the operations of private placement firms and are active in firm affairs. In contrast, in a country like Taiwan where family control is prevalent, individual investors are less likely to hold high levels of ownership and influence major firm decisions. Therefore, many private placements are offered to controlling shareholders and specific conglomerates. This reduces the probability of introducing blockholders to monitor the management through private placements (Hsieh & Wu, 2010).

3. Literature review

3.1. Private placement discounts

In most studies, private placements are issued at a discount (Alli & Thompson, 1993; Barclay et al., 2007; Hertzel, Lemmon, Linck, & Rees, 2002; Hertzel & Smith, 1993; Krishnamurthy et al., 2005; Wruck, 1989; Wruck & Wu, 2009; Wu, 2004). It is widely accepted in the literature that discounts reflect the compensation to investors. Based on the information hypothesis, since investors expend more effort when experiencing difficulties in assessing the firm's value, they correspondingly require higher discounts. For example, Hertzel and Smith (1993) show that the discounts compensate investors for information acquisition costs. Wruck and Wu (2009) suggest that the discounts could also compensate investors for bearing firm-specific risks. Additionally, the certification hypothesis provides an alternative explanation. Because private equity is usually offered to wealthy individuals or institutions that have expertise in valuing firms, when these private investors agree to buy a block of shares (Hertzel & Smith, 1993) or even to buy equity at a premium (Tan et al., 2002), it implies that the issuing firms are worth investing in. Therefore, discounts represent the costs of certification. However, Hertzel et al. (2002) offer a different explanation, which suggests that discounts reflect the investors' assessments of the firm's true (lower) value through observed post-issue stock-price underperformance. Marciukaityte et al. (2005) document that the discounts offered by firms reflect the compensation for purchasing overvalued shares, which is known to investors; therefore, discounts are negatively related to market reactions. Krishnamurthy et al. (2005) argue that when private investors are overoptimistic about a firm's prospects, they purchase the firm's shares at a price that includes the market's valuation on the post-issue stock prices, leading to discounts.

Furthermore, the ownership structure hypothesis offers different views. In private placement studies, the ownership structure hypothesis is often linked with the monitoring argument, suggesting that outside blockholders formed by private placements offer monitoring services and increase firm value. Wruck (1989), Hertzel and Smith (1993) and Wruck and Wu (2009) suggest that the discounts of private placements reflect the compensation to investors for their estimated expert advice and monitoring services.

⁵ Claessens et al. (2000) find that controlling family appoints management in 79.8% of listed firms in Taiwan.

1

In general, when firms form stronger governance relationships with investors through private placements, their price discounts are larger, because the discounts compensate the investors for their participation in the governance process. Barclay et al. (2007) show that discounts reflect the compensation to passive buyers as they seldom participate in the affairs of the firm. This decreases the opportunities for investors to obtain private benefits and indirectly entrenches managers. On the other hand, unlike the case of passive investors entrenching managers, the investors in block transactions willingly buy the shares at a premium for their private benefits of control. The possible reason is that the buyers anticipate receiving benefits that the other shareholders cannot enjoy (e.g., obtaining board seats and participating in firm decision-making) (Barclay & Holderness, 1989).

Finally, in agency theory, managerial self-dealing is a common practice. The hypothesis of managerial self-dealing is that managers appropriate shareholder benefits by buying the firms' shares at low prices or entrenching themselves in private issues (Wu, 2004). Wu and Wang (2005) show that managers may pursue their self- interested wealth, including security benefits (i.e., equity claims) and private benefits (e.g., self-dealing), in addition to the controlling stockholders. When a wedge appears between the control and cash flow rights, issues of private benefits of control arise. In many countries, the managers often hold a small percentage of company shares but remain allied with friendly blockholders, so that they seem to have more power than the dispersed shareholders. In addition, the controlling shareholders may also have the selfish motive of expropriating outside shareholders. In case the interests of minority shareholders are not protected, larger private benefits will be easily obtained (La Porta, Lopez-de-Silanes, & Shleifer, 1999; Shleifer & Vishny, 1997). Wu (2004) shows that if the managers pursue self-dealing, they will buy the shares at low prices and sell them at high prices to transfer the shareholders' wealth to themselves. This is also one of the reasons for private placement discounts.

3.2. Private placement announcement effect

3.2.1. The information effect

When information asymmetry exists in markets, investors regard public issue announcements as negative publicity as they believe that managers seek public equity when the respective firms are overvalued (Miller & Rock, 1985). Private placements could mitigate information asymmetry problems and thus convey the opposite signal to the market. Under the signalling hypothesis, managers convey their valuation of the firms' quality to outside investors through their equity ownership. When managers expect future growth in firms' cash flows, they would retain larger ownership (Leland & Pyle, 1977). Hertzel and Smith (1993) discover that private placements exhibit positive announcement effects that can be attributed to information effects. There exists a positive relationship between the severity of undervaluation and the influence of the information effect. Wu and Wang (2005) note that announcement returns increase with firms' growth expectations. When information asymmetry arises from uncertainty about growth prospects rather than assets in place, the equity issue should generate positive publicity. Additionally, if buyers in private placements are firms' insiders or affiliated investors (i.e., directors, officers, blockholders), the short-run or long-run performance of the issuers will improve. This is because insiders or affiliated investors could have information advantages to adequately assess the firms' value; therefore, they buy private equity shares of undervalued firms (Krishnamurthy et al., 2005; Wruck & Wu, 2009). Hertzel and Rees (1998) observe that the firms' earnings change after private equity issues are positively correlated with firms'

announcement returns, but the underlying riskiness of firms' assets does not change significantly. This result further confirms that a private placement announcement conveys the firm's favourable prospects to the market.

In contrast with underinvestment caused by asymmetric information, Myers and Majluf (1984) propose that financial slack helps alleviate the underinvestment problem. Nonetheless, Jensen and Meckling (1976) argue that managers motivated by self-interest may overinvest and induce firms to grow beyond optimal size. This is an excess cash flow problem because the free cash flow might be wasted on unprofitable projects (Jensen, 1986). Jensen (1986), Myers and Majluf (1984) and Stulz (1990) also agree that following increased liquidity, the firm value will first increase and then decrease. This phenomenon has been widely verified. Even though the majority of the cash flow literature reveal that the market does not favour liquidity accumulation. Hertzel and Smith (1993) indicate that the stock market responds positively when financially distressed firms announce private placements. Although the capital market will penalize excess liquidity, the improved liquidity for non-financial slack firms could elicit a positive response from the stock market. Moreover, LeRoy and Graham (2005) investigate a potential relationship between earnings improvements following private issues and increased liquidity. They observe that when liquidity is enhanced by small firms with growth potential, the announcement abnormal returns will be positive. It is further verified that participation in private issues serves as a medium for investors to convey their private information to the market.

3.2.2. Monitoring, convergence-of-interest and entrenchment

In theory, the blockholders introduced by private placements offer monitoring services and then increase the issuing firms' value (Demsetz & Lehn, 1985; Shleifer & Vishny, 1986). Hertzel and Smith (1993) and Wruck (1989) also suggest that the abnormal returns of private placement reflect the expected monitoring benefits. However, it is unclear whether ownership concentration actually benefits monitoring or only entrenches management. According to the convergence-of-interest hypothesis, since the managers must bear the wealth effects of their decisions, any increase in the managers' ownership helps align their interests with those of outside shareholders, making the costs of divergence from value maximization decrease. Therefore, the value of the firm rises with the managers' ownership (Jensen & Meckling, 1976). Krishnamurthy et al. (2005) also show that if a firm's insiders or affiliated investors participate in private placements, the announcement abnormal returns for the firm will be higher, because the participation helps the managers increase their ownership in the firm. On the other hand, even earlier capital structure theories suggest that an alignment between the objectives of the managers and shareholders is good for the development of a firm. Barclay et al. (2007) argue that capital structure decisions aggravate agency problems because the managers usually place blocks of shares with passive investors to increase their control rights, namely, entrenchment. The result shown in Barclay et al. (2007) is not a special case. Under the entrenchment hypothesis, the managers of firms prefer to sell their shares to friendly investors who will not influence their power. For example, Dann and DeAngelo (1988) find that it is a common technique for managers to block any takeover attempt by privately placing the equity with friendly investors or with themselves. Finally, non-linear arguments offer a comprehensive explanation. Morck, Shleifer, and Vishny (1988) and Wruck (1989) all find that the effect of a change in ownership concentration on firm value is not linear, but varies with the level of ownership. Morck et al. (1988) show that the value of a firm (i.e., Tobin's Q) increases as board ownership rises from 0% to 5%, decreases as it rises from 5% to 25%, and increases again as it rises over 25%. Wruck (1989)

also shows a positive relation between ownership concentration and abnormal returns from private placement when the changes in ownership concentration are in high and low levels and negative in the medium level (between 5% and 25%). The positive correlation reflects the convergence of interests between managers and shareholders, although under some situations the changes in ownership can lead to entrenching managers.

4. Methodology

In this study, we use the multiple regression models to analyze which factors affect the pricing and announcement effects in private placements. Variables are described as follows.

4.1. Dependent variable

4.1.1. Private placement discounts

As in Hertzel and Smith (1993), the discount is measured relative to share price, P_{10} , 10 days after the private placement announcement date⁶ as

$$Discount = \frac{P_{10} - P_0}{P_{10}} \tag{1}$$

If P_{10} is higher than placement price (P_0) , Discount is positive; if P_{10} is lower, Discount is negative. That is, premiums are viewed as negative discounts.

4.1.2. Discount-adjusted abnormal returns

Discount-adjusted abnormal returns ($AR_{\rm adj}$) follows Wruck's (1989) formula.⁷ We adjust the cumulative abnormal returns by taking account of the ratio of shares placed and the discount. The event window is a four-day announcement period from day -3 to day 0 (day 0 is the announcement date of a private placement). Equation is as follows:

$$AR_{\rm adj} = \left[\frac{1}{1-\alpha}\right] [AR] - \left[\frac{\alpha}{1-\alpha}\right] \left[\frac{P_0 - P_b}{P_b}\right] \tag{2}$$

where α is the ratio of shares placed to shares outstanding after the placement, AR is the abnormal stock return over the event window, P_b is the closing price just one day prior to the event window.

We use the market model to measure abnormal returns and estimate betas using the Scholes and Williams (1977) method to correct thin trading.

4.2. Independent variables

Explanatory variables are classified into three categories to explore the information hypothesis, the monitoring hypothesis, and the characteristics of the market, firms and private offerings. The measures of the variables are presented as follows.

4.2.1. Fraction

Fraction denotes the ratio of shares issued to total shares outstanding after an issue. Hertzel and Smith (1993) suggest that when facing profitable investment opportunities but lack sufficient capital, firms can raise the funds they need through private placements to mitigate the underinvestment problem (Myers & Majluf, 1984). When the size of the investment is larger, the firms may place a larger fraction of shares to raise the funds needed. Therefore, the fraction of shares placed reflects the investment opportunities. It is understandable that investment opportunities are more difficult to evaluate than are existing assets; moreover, the information costs may become higher for investors because they may have to expend more time and resources on firm evaluation. From the investors' viewpoint, a larger fraction of shares placed represents higher information costs. From the firms' viewpoint, placing a greater fraction of shares with a long lockup period is practically more difficult. The firms may therefore compensate investors by providing larger discounts (Maynes & Pandes, 2011). Tan et al. (2002) find that the price of placements and the fraction of shares placed exhibit a positive correlation, because the fraction of shares placed reflects the available investment opportunities and affects the earning prospects of the issuing firms. Therefore, a higher fraction means more positive information, and abnormal returns would be higher.

4.2.2. LnProceeds

LnProceeds denotes the natural logarithm of private placement proceeds. When the issue size (issue proceeds) of a firm is larger, more investors will assess the firm. This may reduce the degree of information asymmetry, the possibility of undervaluing the firm, and the costs of information gathering, and therefore the discount will be smaller.

4.2.3. LnMV

Large firms are easily assessed by investors and less likely to be undervalued; the costs of information gathering for investors are therefore lower so price discounts are smaller in large firms. On the other hand, information asymmetry is rather high in small firms, and investors have to incur more costs to assess firm value; the discounts are therefore higher and information effects stronger in small firms. This means that a firm's equity market value is negatively related to abnormal returns (Hertzel & Smith, 1993). Wu and Wang (2005) show that when a small but growing firm raises funds, the announcement abnormal returns will be greater if information asymmetry arises from investment opportunity rather than from assets in place. We use the firm's equity market value as a proxy for firm size. The variable is defined as the natural logarithm of the stock price on the announcement date multiplied by the number of shares outstanding prior to the placement.

4.2.4. BM

BM denotes the year-end ratio of book value of equity to market value of equity prior to the private placement announcement. Empirical studies usually employ BM (book-to-market ratio) as a proxy for growth opportunities (e.g., Barclay & Smith, 1995; Cronqvist & Nilsson, 2005; Krishnaswami, Spindt, & Subramaniam, 1999). Folta and Janney (2004) suggest that a firm with a low BM implies low growth opportunities, but this is more difficult to be valued for investors who will thus obtain a higher discount. Hertzel and Smith (1993) and Tan et al. (2002) also use BM to measure firm value. A lower BM means that a larger fraction of the firm value is from intangible assets, while a higher BM indicates a larger fraction of tangible assets. When the intangible assets of a firm are worth more, information asymmetry for the firm will be more severe, and its value is more likely to be undervalued. Wu, Wang, & Yao

⁶ When measuring the discount, there are two bases used in literature. Krishnamurthy et al. (2005) argue that the aim of computing the discount is to show any benefit from the private issue, so they measure the discount on the price ten days after the announcement. Barclay et al. (2007) argue while negotiating the terms of the private placement, the traders should have considered the impact of the announcement on the market price, thus they use the price one day after the announcement to obtain the discount. The others using price after announcement to measure premium/discount include Hertzel et al. (2002), Wu (2004), Lee and Wu (2009). Instead, Alli and Thompson (1993), Tan et al. (2002), Anderson et al. (2006), LeRoy and Graham (2005) and Wu et al. (2005) use preannouncement price to obtain premium/discount. Wruck and Wu (2009) use both methods.

 $^{^{7}}$ Although the measure $AR_{\rm adj}$ was questioned by Wu et al. (2005), it is more appropriate in exploring information content of private placement announcement. Therefore, we still use $AR_{\rm adj}$ as the dependent variable of announcement return regressions.

c

(2005) extend Wu and Wang's (2005) theory, and proposes another view: if the lower market-to-book ratio (MV/BV) of a firm is due to its lower value of assets in place rather than future investments, it means that there is sufficient uncertainty over growth, and the private placement announcement effect may thus be better.⁸ Conversely, if the lower MV/BV results from bad past investments, the announcement effect may be worse.

4.2.5. CAR(-59, -2)

As in Chen, Ho, Lee, and Yeo (2002b), we use cumulative abnormal returns over the period beginning 59 days and ending 2 days prior to the private placement announcement. The purpose is to control mean reversion in returns.

4.2.6. Purpose

While studying the issue on moral hazard, 9 Cronqvist and Nilsson (2005) find that in contrast to rights offerings, the sample firms often develop a new product-market relationship with investors, such as production technology, R&D, distribution systems and marketing, at the time of private offerings. Barclay et al. (2007) divide private investors into three types: active, ¹⁰ managerial, and passive. If a purchaser becomes an active investor in firm affairs following the placement, both the short- and long-run stock abnormal returns will be positive. This clearly shows that raising capital with a strategic purpose conveys a positive signal about private placements. Jang et al. (2010) study the private placements in Taiwan and find that when the placements have a financial purpose, the relation between changes in ownership concentration and stock prices is positive, implying that the insiders believe the firm is underpriced or there are benefits of alignment between the managers and investors. If the placements have a strategic purpose, when the extent of ownership concentration decreases due to increased liquidity or increased outside blockholder monitoring, the stock prices will rise.11

If the private placements have a strategic purpose, we define *Purpose* as equal to 1; if the purpose is financial, we define *Purpose* as equal to 0. Strategic purposes include using the issue proceeds for mergers and acquisitions (M&As), backdoor listing and ensuring the rights to operate. Financial purpose announcements include using the proceeds for working capital, capital expenditure, reinvestment or improving the financial structure. The data are from MOPS in Taiwan, 12 but due to the rough information, we use the data collected by the InfoWinner Plus database as auxiliary to explore the true intentions in private placement firms.

4.2.7. Distress

A firm is regarded as under financial distress if it experiences two consecutive years of negative earnings prior to the private placement announcement, or as provided by the *Taiwan Economic Journal* database, if it faces serious events that threaten its operations, such as bankruptcy or restructuring. If a firm is under financial distress, we define *Distress* as equal to 1; otherwise, *Distress* is equal to 0.

In Krishnamurthy et al. (2005), this variable (Distress) is used to indicate whether the public issue of shares is a feasible choice for firms. In private placements, adverse selection problems¹³ are mitigated through the participation of insiders and informed investors. Hence, even if financially distressed firms face no information asymmetry and related moral hazard problems, private placements are often the only viable financing channel. This suggests that the information argument applies only to firms that have the ability to issue equity through both the private and public equity markets. On the other hand, by offering a controlling interest, private placements may be able to attract professional investors who would help the firms tide over financial issues. Similarly, even if such firms face no information asymmetry and related moral hazard problems, they still need to rely on private placements. Therefore, while studying the effects of certification by investors in information theory, it would be more meaningful to study firms that are not financially distressed.

4.2.8. Year

Although a firm's financial condition influences the method it adopts to attract capital (Hertzel & Smith, 1993; Krishnamurthy et al., 2005), the stock market will respond to news at different speeds during boom and recession periods (Antweiler & Frank, 2006), thereby inducing the firms to adopt different financing policies under various market conditions. For example, private placements may be influenced by bull or bear markets. Since our study covers the financial crisis period, after observing the stock price movements in Taiwan, we define *Year* as follows: if the placement is announced between January 2001 and July 2007, *Year* is equal to 0; if the announcement is between August 2007 and December 2008 (during the financial crisis period), *Year* is equal to 1.14

4.2.9. Exchange

Spiess and Affleck-Graves (1995) find that long-run underperformance after seasoned offerings can be more severe for firms listed in NASDAQ. In another paper on insider trading in the OTC market, Lin and Howe (1990) show that firms in the OTC market are relatively small, and that smaller firms may be covered by fewer institutional investors and stock market analysts, leading to a larger degree of information asymmetry. Therefore, we suggest that discriminating the exchanges to which the private placement firms belong should have special meaning. If the private equity issuer is listed on Taiwan Stock Exchange (TSE), *Exchange* is equal to 1; if the issuer is listed on Gre-Tai Securities Market (OTC), *Exchange* is equal to 0.

⁸ Wu et al. (2005) divide MV/BV into two parts: A is the market value of assets-in-place and B is the expected NPV of future investments. Then MV/BV can be rewritten for (A+B)/BV. A Low (A+B)/BV may not necessarily result from poor growth prospects (B) but from bad past investments (A), leading to a turning point for issuers if uncertainty over growth is large enough.

⁹ For example, Klein et al. (1978) have pointed out that when firms cooperate with other firms for some investment projects, any party may break contracts for self interests. This is one of moral hazard risk.

¹⁰ Barclay et al. (2007) define active investors as those interacting with issuing firms after placements within 2 years (e.g., joint research, combined marketing and the purchasers' joining the board of the issuing firm).

¹¹ In financial literature, the use of private issue proceeds is often classified into capital expenditure and working capital (e.g. Tan et al., 2002; Anderson et al., 2006). Capital expenditure implies that the firm has good investment opportunities to enhance earnings, so it represents a positive signal. Based on robustness, this study also classifies the use of issue proceeds as (1) for capital expenditure, business expansion and strategic alliances, and (2) for repayment of debt, improving financial structure or financing working capital needs. Because the results are similar to those of the classification by strategic and financial purposes, for brevity, the results are not presented in this article.

¹² Available at http://mops.twse.com.tw.

¹³ For example, when information asymmetry problems exist, firms can not be valued properly. Undervalued firms face valuable investment opportunities but lack sufficient capital will give up raising capital in public market (Myers & Majluf, 1984). This is one of adverse selection problems.

¹⁴ For estimating the occurrence of global financial crisis, most people use the Lehman default (September 15, 2008) as the watershed. Just as in "The Crisis" announced in 2010 by Greenspan, American former chairman of FED, the financial crisis erupted from the subprime mortgage event (summer 2007). The stock prices also fell from August 2007 in Taiwan; so we choose this period to define the Year.

4.2.10. Managers

Managers include directors, supervisors and senior management. If the managers of firms participate in private placements, Managers is defined as equal to 1; if no managers participate, Managers is equal to 0. According to the information hypothesis, managers should be charged lower information acquisition costs; therefore, increased managerial holdings can be associated with smaller discounts and positive announcement effects because of signalling undervaluation. On the contrary, owing to managerial self-dealing or additional compensation, 15 the managers are likely to require greater discounts, hurting the interests of the non-participating shareholders. Owing to the interaction between the two kinds of conflicting incentives of managers, sales to managers are viewed as a less positive signal than sales to outsiders (Hertzel & Smith, 1993). Additionally, the managers and other affiliated investors of a firm may be too optimistic about the issuer's prospects (Heaton, 2002; Malmendier & Tate, 2005; Wong & Zhang, 2009), whereas outside investors are more objective and expend more effort to obtain better information. Thus, Wruck and Wu (2009) argue that under certain circumstances, placements to outside investors convey more positive information.

4.2.11. Non-manager affiliated

To emphasize the importance of investor identification in studying private placements, Krishnamurthy et al. (2005) divide the investors into affiliated and non-affiliated categories. They cite the argument of Leland and Pyle (1977) and Barclay et al. (2007) to show that the affiliated investors acquire information with lower costs and better understand the prospects and the true value of firms. Therefore, the participation of affiliated investors could be regarded as a certification of firm quality. However, even if the affiliated investors value the firm negatively, they can still be compensated by discounts; therefore, an increase in affiliated ownership may render the insiders more entrenched and increase agency problems. Lin and Howe (1990) find that once the insiders (directors, officers and blockholders holding over 5% shares) use their valuable information to time their transactions, they will easily benefit more than the unaffiliated shareholders. Therefore, investor identification has always been one important argument in the literature.

When the affiliated investors are not managers, we define *Non-manager affiliated* as equal to 1; otherwise, *Non-manager affiliated* is equal to 0. Following the definition of Krishnamurthy et al. (2005), with a slight modification based on media and proxy statements, the affiliated investors include the top management, managers' relatives, attorneys or the correspondent bank of the firm, current block shareholders of the firm (holding more than 5% shares), strategic alliance institutions, and the companies having product market agreements with the firm.

4.2.12. Single

When a single investor participates in placements, we define *Single* as equal to 1; otherwise, *Single* is equal to 0. For placements with single investors, the discounts are expected to be smaller, because single investors would be willing to pay a higher price to obtain control rights, or the information costs are smaller than those in placements with multiple investors. Hertzel and Smith (1993) argue that if the control premium partially offsets the information costs, it will result in smaller discounts and greater abnormal returns.

4.2.13. Ownership and △Ownership

Ownership is defined as the percentage holding of all the managers and non-management investors with 5% or greater ownership.△Ownership is the difference between the ownership concentration before the private placement announcement and after the announcement. Researchers usually explore the monitoring motives for private placements in two aspects: ownership concentration and trading restrictions (Wu, 2004). Wruck (1989) proposes that an increase in ownership concentration can improve the monitoring mechanism. Nevertheless, if most of the ownership is held by passive investors, the monitoring mechanism may not necessarily improve. This is supported by Barclay et al. (2007), who suggest that most private investors are passive in firm affairs, so the discounts reflect the compensation for entrenching managers. Wu (2004) makes a deep study of whether managers engage in self-dealing or not. In theory, monitoring is more beneficial to shareholders with large initial holdings. To encourage the investors with small initial holdings to buy private equity shares, the issuing firms must offer large price discounts to compensate for liquidity restrictions. This may lead to price discounts for shareholders with large initial holdings being smaller. Nonetheless, Wu's (2004) empirical results do not support this argument and are more consistent with the managerial self-dealing hypothesis.

In summary, the information hypothesis variables include Fraction, LnProceeds, LnMV, BM, Distress, and Single. The monitoring hypothesis variables include Managers, Non-manager affiliated, Single, Ownership and \triangle Ownership. The variables for the market, characteristics of firms and placements include CAR (-59, -2), Purpose, Distress, Year and Exchange. In addition, since Single and \triangle Ownership represent similar meanings, Hertzel and Smith (1993) suggest that Single is a discrete variable and hence does not reflect the size of the placement; \triangle Ownership is a continuous variable that reflects the size of the placement, but is not easily available and exhibits measurement errors (if ownership changes result from non-placement reasons). For robustness, we use both the variables in regression analysis by turns.

5. Empirical analysis

5.1. Sample construction

We use a sample of private placements in conjunction with common stock that were announced between January 2002 and December 2008 by firms listed on Taiwan Stock Exchange (TSE) and Gre-Tai Securities Market (OTC). Data are collected from the Market Observation Post System (MOPS)¹⁶ in Taiwan, along with the annual reports, InfoWinner Plus database. Besides, directorship, financial and return data are obtained from the *Taiwan Economic Journal* database.

We exclude observations that contain other events during the private placement announcement period, which concern capital reduction, dividends, equity repurchases and issuance announcements of other securities (preferred stock, GDR and corporate bonds), to avoid confounding effects. In addition, we eliminate

¹⁵ For example, compensate managers for less diversified risk (Wu, 2004).

¹⁶ The MOPS, established by TSE and OTC in June 2002, is a convenient platform for listed companies to place their public information and for investors to search for information. However, information on private placements until 2005 is not offered directly on the MOPS. We collected private placement data by filtering all the important information offered by the firms on the MOPS from 2002 to 2004, and then verifying the news released in the newspapers around the private placement announcement date. We further verified the annual reports to ensure the correctness of the data. Although the information offered on the MOPS from 2005 is more complete, it is still necessary to collect information such as the identities of investors and the purpose of private placements from the newspapers and annual reports.

H.-C. Liang, W.-Y. Jang / The Quarterly Review of Economics and Finance xxx (2013) xxx-xxx

 Table 1

 Sample characteristics of equity private placements.

Variable		N		Mean		Median		
Panel A: continuous vario	ables							
Fraction (%)		326		20.678		16.490		
Proceeds – US\$ million		326		12.490		3.487		
MV - US\$ million		326		121.075		20.744		
BM		317		0.928		0.788		
Ownership (%)		326		32.389		29.420		
△Ownership (%)		326		3.158		0.862		
Classification	N		Percentage (%	(S) Classifi	cation		N	Percentage (%)
Panel B: dummy variable	?S							
Single	65		20	TSE			129	40
Multiple	258		80	OTC			197	60
2–5		104	32					
6-12		92	29	During	the financial crisis	s period	111	34
More than 13		62	19	Not du	ring the financial o	risis period	215	66
Affiliated	263		81	Strateg	ic purpose		29	9
Managers		224	69	Financi	al purpose		297	91
Non-manager affiliate	ed	39	12					
Nonaffiliated	60		19	Under f	inancial distress		155	48
				Not une	der financial distre	ess	171	52
Fiscal year	2002	2003	2004	2005	2006	2007	2008	Summary
Panel C: distribution of p	rivate placements by	year						
Frequency	3	15	19	48	81	89	71	326
Percentage (%)	0.92	4.60	5.83	14.72	24.85	27.30	21.78	100.00

Note: The table presents the sample characteristics of the private placement and the firms placing equity privately during the period 2002–2008. Panel A reports continuous variables, and Panel B reports dummy variables. Fraction is the ratio of shares issued to total shares outstanding after an issue. LnProceeds is the natural logarithm of private placement proceeds. LnMV is the natural logarithm of the stock price on the announcement date multiplied by the number of shares outstanding prior to the placement. BM is the ratio of book value of equity to market value of equity. Ownership is the percentage holding of all the managers and non-management investors with 5% or greater ownership. \(\triangle Ownership \) is the difference between the ownership concentration before the private placement announcement and after the announcement. Single investor means only a single investor participates in placements. Affiliated investors include Managers (directors, supervisors and senior management) and Non-manager affiliated (managers' relatives, attorneys or the correspondent bank of the firm, current block shareholders of the firm, strategic alliance institutions, and the companies having product market agreements with the firm). (Not) during the financial crisis period means the placement is announced between August 2007 and December 2008 (between January 2001 and July 2007). Strategic purposes include using the issue proceeds for M&As, backdoor listing and ensuring the rights to operate. Financial purposes include using the proceeds for working capital, capital expenditure, reinvestment or improving the financial structure. A firm is regarded as under financial distress if it experiences two consecutive years of negative earnings prior to the private placement announcement, or as provided by the Taiwan Economic Journal database, if it faces serious events that threaten its operations.

the private placements not issued within one year after the initial announcement¹⁷ and those without complete information regarding the variables of interest. Financial institutions are also excluded from the sample by convention, because firms in this industry have special operating and financial structures.

Our initial sample consists of 439 observations, of these, 113 observations not meeting requirements are eliminated, ¹⁸ leaving us a final sample of 326 observations, ¹⁹ including 193 firms (some firms conduct more than one placement).

5.2. Descriptive statistics

Table 1 provides the descriptive statistics to describe the sample characteristics. Panel A shows results for the continuous variables. On average, the ratio of shares issued to total shares outstanding after the issue is 20.7% (median 16.5%). The average proceeds raised

from private placements in our sample is \$12 million (median \$3 million). Of our sample firms, the average equity market value is \$121 million (median \$21 million), and the average book-to-market ratio is 0.928 (median 0.788). Total percentage holdings by managers and 5% or greater non-management averages 32.4% (median 29.4%) before private placements, and the average change in total percentage holdings is 3.2% (median 0.9%).

Panel B reports the results for the dummy variables, showing that about 69% (81%) of private placements are placed with managers (affiliated investors); this is similar to the evidence in Wruck and Wu (2009). In Wruck and Wu (2009), the percentage of private placements sold to investors who have a pre-issue relationship (e.g., managers, business partners and prior block holders) with the issuers is also large (64%). We conjecture that the high proportion of private placements involving insiders/managers may be related to the market characteristics in Taiwan. When a family controlled firm with many family members on the board of directors is likely to be under financial distress, and needs to raise funds but is not willing to lose control rights, the effective way is to offer their private placement shares to the family members (insiders and managers). The explanation corroborates that a rather large number of our sample firms have faced financial distress. In Panel B, we may find 48% of the sample firms experience two consecutive years of negative earnings prior to the announcement or financial distress news releases

Besides, our sample shows that about 20% of private placements are sold to single investors and 32% are sold to fewer than 6 (2–5) investors. Earlier literature also finds that most private equity

¹⁷ Since the firms are permitted to divide the private equity shares into parts to issue within one year, the earliest pricing announcement date by the same private placement is used as criteria for deciding the sample.

¹⁸ We exclude 113 observations in total. Sample reduction reasons include: simultaneous announcements of capital reduction, dividends or equity repurchases (13 observations), unavailable announcement date data (11 observations), a placement not completed after announcement (53 observations), insufficient stock-price data (30 observations) and outliers (6 observations).

¹⁹ Of these, 201 private placements are from the computer-related industry, 47 from construction, 14 from steel products, 10 from electric machinery, 6 from textiles, and the rest are from other 9 industries.

 Table 2

 Price discounts and announcement period returns in equity private placements.

			N		Mean (%)		Median (%)		
Panel A: total sample									
Discount 326					9.554 (2.970)***		22.048 (8.316)***		
CAR(-3, 0)	CAR(-3,0) 302				1.419 (2.730)***		0.536 (2.572)**		
CAR (-29, 10) 302					7.153 (3.921)***	6.215 (3.990)***			
Discount-adjusted CAR					4.680 (5.618)***				
	TSE			ОТС			Difference		
	N	Mean (%)	Median (%)	N	Mean (%)	Median (%)	Mean (%)	Median (%)	
Panel B: subsamples by e	xchange								
Discount	129	6.653	20.000***	197	11.454***	23.491***	-4.802	-3.491^{*}	
		(1.249)	(4.639)		(2.844)	(6.886)	(-0.729)	(1.765)	
CAR(-3, 0)	121	2.122**	1.217**	181	0.948	0.232	1.174	0.984	
		(2.599)	(2.442)		(1.409)	(1.336)	(1.107)	(1.027)	
CAR (-29, 10)	121	6.686**	6.294**	181	7.466***	6.137***	-0.781	0.158	
		(2.537)	(2.473)		(3.001)	(3.128)	(-0.209)	(0.128)	
Discount-adjusted	120	3.674	4.345***	177	2.829	4.807***	0.845	-0.461	
CAR(-3 to 0)		(1.527)	(4.362)		(1.264)	(3.773)	(0.251)	(0.105)	

Note: The table presents price discounts and announcement period returns in private placements. The discount is measured relative to share price 10 days after the announcement. CAR is the average cumulative abnormal returns over different intervals around the announcement. Discount-adjusted CAR= $[1/(1-\alpha)]$ [AR] – $[\alpha/(1-\alpha)]$ [AR] – $[\alpha/(1-\alpha)]$] [AR] – $[\alpha/(1-\alpha)]$] [AR] – $[\alpha/(1-\alpha)]$] [AR] – $[\alpha/(1-\alpha)]$] where α is the ratio of shares placed to shares outstanding after the placement, AR is the abnormal stock return over the event window, R is the closing price just one day prior to the event window. A t-test (Wilcoxon signed rank/Mann–Whitney R] is used to test the null hypothesis that the mean (median/median of the differences) is zero. Test statistics are reported in parentheses.

- * Indicates significance at the 10% level.
- ** Indicates significance at the 5% level.
- *** Indicates significance at the 1% level.

shares are placed with a selective number of investors (Wruck, 1989; Wruck & Wu, 2009). The OTC firms conduct more private placements (60%). The proceeds from 9% of private placements are used for strategic purpose. Although the overlap between our study period and the financial crisis period is only 17 months, 34% of the private placements are placed during the overlap period; Panel C reports the time distribution of the sample further. Since private placements were permitted from 2001 in Taiwan, the number of placements is not large during the first few years, although growing annually. Chou, Gombola, & Liu (2009) and Mathew (2002) find that the number of private issues in a single year is largest when the stock prices reached high levels in the US and Hong Kong, respectively. In Taiwan, stock prices were seen to rise annually during the study period but begin falling during the second half of 2007. This supports the view that firms issue equity to exploit their overpriced stock. Additionally, since the private placement is a new financing instrument in Taiwan, we conjecture that the growth in the number of placements may also be due to the market gradually becoming familiar with the instrument. Finally, the number of private placements is seen to reduce in 2008, for which the financial crisis may be a possible reason, and the managements have become conservative in financing activity due to greater systematic

5.3. Price discounts and announcement period returns in private placements

Table 2 reports price discounts and announcement period returns in private placements. In Panel A, the average discount in the sample is 9.554%, significant at the 1% level. When the sample is classified by listed exchange (Panel B), the average discount is smaller for the TSE firms $(6.653\%^{20})$ than for the OTC firms (11.454%). As for announcement effects, the four-day CAR in the window (-3,0) is shown as 1.419%, and discount-adjusted

abnormal return is 3.170%, both significant at the 5% level and higher for the TSE firms. Due to the potential leakage of information, Hertzel and Smith (1993) suggest using a longer event window, such as the $(-29,\ 10)$ window. In Table 2, the CAR in the window $(-29,\ 10)$ is significantly positive but higher for the OTC firms. In general, positive abnormal returns prior to the announcement day can be explained as the firms issuing private equity during the periods of stock-price run-ups. Information leakage is another explanation for private placements being often announced at the date of continuing or completed private placements (Tan et al., 2002; Wu et al., 2005), resulting in those with superior information making profits through speculative trading.

5.4. The pricing of private placements

5.4.1. Regression analysis of private placement discounts

In this study, we use the multiple regression models to analyze which factors affect the magnitude of the discounts in private placements. Table 3 reports the cross-sectional regression of private placement discounts, showing the relationship between the independent variables and discounts for the sample firms. In all models of the full sample, the coefficients on Fraction are positive and significant at 1% level. This indicates that the discounts reflect the information search costs. When the fraction of placements is higher, the growth opportunities of firms are relatively more difficult to evaluate, leading to increases in the costs of information search borne by the investors, who would thus expect larger discounts as compensation. Moreover, it is also possible that the investors may presume that the stock price of a firm that places a larger fraction of shares may decline, and thus, require larger discounts as compensation. The coefficients on LnMV are significantly positive, which is inconsistent with the view that information asymmetry is more severe for small firms, and so the investors would expect larger discounts to compensate their valuation costs. According to Hertzel and Smith (1993), large firms need to be more closely monitored, and so their greater discounts reflect the compensation for monitoring costs. The significantly negative coefficients on Exchange may be due to better information

 $^{^{20}}$ Though the average discount is only 6.63%, 65% of TSE firms have a discount higher than 6.63%.

H.-C. Liang, W.-Y. Jang / The Quarterly Review of Economics and Finance xxx (2013) xxx-xxx

Table 3Cross-sectional regression of private placement discounts.

	Full sample			TSE			OTC		
	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3
Intercept	-4.082***	-3.913***	-4.399***	-3.718**	-3.325*	-3.928**	-3.562***	-3.446***	-3.853***
	(0.001)	(0.001)	(0.000)	(0.049)	(0.059)	(0.035)	(0.000)	(0.002)	(0.000)
Fraction	1.952***	2.070***	1.963***	0.684	0.954	0.666	3.346***	3.431***	3.241***
	(0.001)	(0.001)	(0.001)	(0.229)	(0.129)	(0.245)	(0.000)	(0.000)	(0.000)
LnProceeds	-0.137	-0.126	-0.144	0.029	0.054	0.014	-0.401**	-0.402^{**}	-0.387**
	(0.176)	(0.228)	(0.121)	(0.732)	(0.561)	(0.865)	(0.012)	(0.010)	(0.019)
LnMV	0.314***	0.299***	0.335***	0.146**	0.109	0.176**	0.514***	0.511***	0.515***
	(0.000)	(0.000)	(0.000)	(0.048)	(0.142)	(0.020)	(0.000)	(0.000)	(0.000)
BM	0.097*	0.083	0.084**	0.116	0.105	0.054	0.090*	0.079	0.123**
	(0.076)	(0.107)	(0.035)	(0.143)	(0.150)	(0.274)	(0.084)	(0.153)	(0.026)
CAR(-59, -2)	-0.141*	-0.130*	-0.102	-0.006	-0.038	0.052	-0.150**	-0.139**	-0.148**
e. m (35, 2)	(0.055)	(0.068)	(0.128)	(0.963)	(0.752)	(0.678)	(0.019)	(0.035)	(0.022)
Distress	0.032	0.039	-0.049	0.062	0.119	-0.075	-0.013	-0.015	-0.030
Distress	(0.718)	(0.665)	(0.509)	(0.683)	(0.477)	(0.520)	(0.876)	(0.849)	(0.715)
Year	-0.084	-0.068	-0.051	-0.177	-0.129	-0.159	-0.013	-0.008	0.012
rear	(0.245)	(0.309)	(0.450)	(0.191)	(0.260)	(0.189)	(0.833)	(0.903)	(0.867)
Purpose	-0.103	-0.184**	-0.012	-0.036	-0.188*	0.019	-0.087	-0.122	-0.034
Turpose	(0.221)	(0.022)	(0.904)	(0.815)	(0.099)	(0.913)	(0.394)	(0.210)	(0.734)
Exchange	-0.197**	-0.209**	-0.153**	(0.013)	(0.033)	(0.313)	(0.554)	(0.210)	(0.754)
Exchange	(0.039)	(0.035)	(0.037)						
Managers	-0.100	-0.047	-0.049	-0.039	0.033	-0.023	-0.090	-0.062	-0.074
Managers	(0.268)	(0.518)	(0.568)	(0.784)	(0.783)	(0.869)	(0.371)	(0.450)	(0.508)
Non-manager affiliated	-0.157	(0.518) -0.124	-0.099	-0.001	0.033	-0.007	(0.371) -0.227	-0.196	-0.201
Non-manager animated									
6: 1	(0.108)	(0.197)	(0.342)	(0.992)	(0.817)	(0.960)	(0.101)	(0.134)	(0.147)
Single	-0.180*		-0.452**	-0.283		-0.699*	-0.080		-0.231
0 1:	(0.090)	0.000*	(0.038)	(0.121)	0.00=**	(0.094)	(0.434)	0.040	(0.144)
Ownership		-0.399*			-0.807**			-0.212	
		(0.086)			(0.046)			(0.396)	
△Ownership		-0.496*			-1.222 [*]			-0.246	
		(0.074)			(0.068)			(0.310)	
Single*Fraction			0.678			0.557			0.705
			(0.227)			(0.546)			(0.273)
Single*BM			-0.018			0.059			-0.116
			(0.902)			(0.818)			(0.450)
Single*Distress			0.528***			0.663**			0.217
			(0.009)			(0.033)			(0.290)
Single*Exchange			-0.159						
			(0.401)						
N	307	307	307	122	122	122	185	185	185
Adjusted R ²	0.197	0.195	0.231	0.145	0.164	0.192	0.334	0.332	0.334
Aujusteu K-	0.197	0.195	0.231	0.145	0.104	0.192	0.334	0.332	0.334

Note: The dependent variable is the private placement discount measured relative to share price 10 days after the announcement. Fraction is the ratio of shares issued to total shares outstanding after an issue. LnProceeds is the natural logarithm of private placement proceeds. LnMV is the natural logarithm of the stock price on the announcement date multiplied by the number of shares outstanding prior to the placement. BM is the ratio of book value of equity to market value of equity. CAR (−59, −2) is the cumulative abnormal returns over the period beginning 59 days and ending 2 days prior to the private placement announcement. Distress equals to 1 if a firm is under financial distress; otherwise 0. A firm is regarded as under financial distress if it experiences two consecutive years of negative earnings prior to the private placement announcement, or as provided by the Taiwan Economic Journal database, if it faces serious events that threaten its operations. Year equals to 1 if the placement is announced between August 2007 and December 2008, and 0 if the announcement is between January 2001 and July 2007. Purpose equals to 1 if the private placements have a strategic purpose and 0 if the purpose is financial. Exchange equals to 1 if the private equity issuer is listed on TSE and 0 if the issuer is listed on OTC. Managers equals to 1 if the managers of firms participate in private placements; otherwise 0. Non-manager affiliated equals to 1 if the affiliated investors are not managers; otherwise 0. The affiliated investors include Managers (directors, supervisors and senior management) and Non-manager affiliated (managers' relatives, attorneys or the correspondent bank of the firm, current block shareholders of the firm, strategic alliance institutions, and the companies having product market agreements with the firm). Single equals to 1 if a single investor participates in placements; otherwise 0. Ownership is the percentage holding of all the managers and non-management investors with 5% or greater ownershi

- * Indicate significance at the 10% level.
- ** Indicate significance at the 5% level.
- *** Indicate significance at the 1% level.

10

disclosures or liquidity²¹ by the TSE firms, leading to smaller discounts, in line with the information asymmetry hypothesis.

The coefficients on CAR (-59, -2) are significantly negative in both the full sample and the OTC subsample, but not significant in the TSE subsample. What is the implication of stock-price run-ups before private placements that cause OTC firms to raise funds with smaller discounts? As a study in Taiwan, He (2009) shows that the OTC firms release more positive signals about private placements during the announcement period, whereas TSE firms release both positive and negative signals almost evenly (the observation window is during the 2-month period before and after the announcement date). Therefore, we conjecture that the issuing firms may intentionally release positive signals during the issuing period to create opportunity windows for themselves.

When we replace Single with $\triangle Ownership^{22}$ in Model 2, the variable "Purpose" becomes significant in the full sample and the TSE subsample, suggesting that when the purpose of private placements is strategic consideration, the discount is smaller than for financial consideration. The coefficient on Ownership is significantly negative in the full sample and the TSE subsample, implying that when the insider and shareholder interests are aligned, the discounts are smaller to reduce wealth transfers to new shareholders. Besides, since the information search costs are smaller for insiders, the discounts are smaller, which is consistent with the argument of Hertzel and Smith (1993). Moreover, the significantly negative coefficient on $\triangle Ownership$ means that the insiders purchase private placement shares at higher prices to gain control benefits. This is consistent with the results for the variable "Single" shown below.

On the other hand, some variables can be used to explain both the information and monitoring effects; we therefore study the interaction of the Single variable with the variables related to the information hypothesis to determine the differences. Hertzel and Smith (1993) suggest that the likelihood of monitoring will increase when the ownership of private equity shares is concentrated. Therefore, if the monitoring hypothesis holds, we can test whether the magnitude of the discount is related to the expected monitoring cost by the interaction terms. If the relation shows, the expected signs of the interaction terms should be the same as the corresponding variables. The results of all the samples are shown in Model 3 (due to multicollinearity, we eliminate two interaction terms: Single × LnProceeds and Single × LnMV). At first, the coefficients on Single are significantly negative in both the full sample and the TSE subsample. If this is according to the information hypothesis, the negative coefficient on Single means that the smaller discount in private placements participated by single investors reflects the information cost savings (if by more participants, the informing cost relatively increases). If it is based on the monitoring hypothesis, the smaller discount implies that the control premium by block holdings offsets the expected monitoring cost in part, which is consistent with Hertzel and Smith's (1993) findings. Wayne and Hailu (1991) also find that investors pay premiums for control or voting rights in block transactions. Are the results of the sample a consequence of information advantages or control benefits? After analyzing the sample data, we find that under some circumstances, private equity shares are purchased at a premium by single investors. For example, 43% (49%) of private placements are offered at a premium in the TSE firms (financially non-distressed firms). If this is based only on the information advantage argument, it does not explain why investors are willing to purchase shares at a premium. Thus, the control premium explanation appears more adequate here. For further confirmation, see Section 5.5, where we show the regression of announcement abnormal returns.

The coefficient on the interaction term between *Single* and *Distress* is significantly positive, showing that when the issuing firms face financial distress, they must compensate the investors with discounts for increased monitoring, such as by introducing blockholders to participate in firm affairs or bringing new management skills to improve the operating performance (in line with the monitoring hypothesis). It is also possible that the financially distressed firms cannot easily shop around with multiple potential buyers who would need the time to perform due diligence before making an investment in a financially distressed firm (in line with the information hypothesis). However, why does not the phenomenon appear in the OTC firms? We conjecture that the fund-raising purpose of the OTC firms facing financial distress is to survive or to obtain operating rights, and the investors no longer care about the prices of private equity shares.

If classified by listed exchanges, information asymmetry definitely influences the OTC firms more than the TSE firms. The coefficients on *Fraction*, *LnProceeds* and *BM* are significant in the OTC subsample but insignificant in the TSE subsample. The significantly negative coefficients on *LnProceeds* indicate that information acquisition has economies of scale, and so the discounts are smaller for larger placements. Therefore, we conjecture that the information hypothesis applies appropriately to the OTC firms.

5.4.2. Managerial self-dealing

Because managers are self-interested in gaining control benefits, this study determines whether they are likely to engage in self-dealing. Wu (2004) shows that managers tend to purchase private equity shares at large discounts when they possess small original holdings; therefore, this study uses Wu's (2004) procedure to test the managerial self-dealing hypothesis. First, the means are compared in Panel A of Table 4, in which Rows 1 and 2 show the discounts for private placements involving managers and for those of non-manager investors. The discounts for private placements involving managers are less in both the full sample and the OTC subsample, but more in the TSE subsample, in which the discounts are greater if managers are involved in private placements. However, the difference in average discounts is insignificant between private placements involving managers or non-manager investors. In Rows 4 and 5, managerial buyers are classified into those with small holdings or with big holdings based on whether their original ownership is below or above their median ownership. The mean discount is 18.180% (median 26.603%) for managers with small holdings and -0.396% (median 18.935%) for managers with big holdings. The mean and median discounts are significantly different between these two categories at the 5% level. We also find the same results for the TSE subsample but not for the OTC subsample. The results indicate that managers in TSE firms are likely to engage in self-dealing.

Panel B shows the different shareholding percentages of managers in firms with small managerial ownership and those in big managerial ownership. For the full sample, the mean shareholding percentage of managers is 12.74% (median 12.91%) for firms with small managerial ownership and 33.63% (median 30.87%) for firms

²¹ When conducting SEOs, the firms having less liquid stocks generally suffer great flotation costs (Hung et al., 2012). For example, Corwin (2003) finds that underpricing is negatively related to stock liquidity, and Butler et al. (2005) show that investment banks charge significantly greater fees on firms with illiquid stocks. Alternatively, Maynes and Pandes (2011) find that share turnover (a common proxy for liquidity) is positively related to private placement discounts in Canada. We have investigated liquidity in TSE and OTC subsamples. Liquidity is larger in TSE than in OTC-offering firms, but including the liquidity measures does not improve the explanation power of our regression models. Since there are mixed results in the liquidity impacts on equity offerings, we thus take a conservative attitude to relevant conjecture and leave the issue for further research.

²² Because Single and △Ownership represent similar meanings, we use both the variables in our regression analysis by turns.

Table 4 Managerial self-dealing analysis.

	Full sample			TSE			OTC			
	N	Mean (%)	Median (%)	N	Mean (%)	Median (%)	N	Mean (%)	Median (%)	
Panel A: private placement discour	nts									
Managers	224	8.892	22.631	86	6.849	20.593	138	10.166	23.328	
Non-managers	99	10.891	20.732	43	6.259	18.182	56	14.447	26.596	
Difference (test statistics)		-0.283	0.085		0.052	0.969		-0.474	0.919	
Managers with small holdings	112	18.180	26.603	43	23.599	30.233	69	16.433	23.333	
Managers with big holdings	112	-0.396	18.935	43	-9.900	13.597	69	3.899	22.764	
Difference (test statistics)		2.300**	2.713***		2.344**	3.429***		1.307	0.969	
	Full sar	Full sample			TSE			OTC		
	N	Mean (%)	Median (%)		Mean (%)	Median (%)	N	Mean (%)	Median (%	
Panel B: shareholding percentage of	of managers									
Managers with small holdings	112	12.74	12.91	43	10.26	10.24	69	14.74	15.00	
Managers with big holdings	112	33.63	30.87	43	31.67	28.72	69	34.40	33.29	
Difference (test statistics)		-19.149***	12.932***		-10.230^{***}	7.981***		-15.627***	10.135***	
		Full sam	ple		TSE			OTC		
		Coefficient P-value			Coefficie	nt P-value		Coefficient P-value		
Panel C: regression of private place	ement discou									
Intercept			**(0.001)		-3.422^{*}			-3.514*** (0.001)		
Fraction		1.903***	` '		0.318 (0.	,		3.378*** (0.000)		
LnProceeds		-0.121			0.080 (0.	,		-0.404 (0.010)		
LnMV		0.301***			0.089 (0.			0.514*** (0.000)		
BM		0.081 (0			0.079 (0.	,		0.074 (0.173)		
CAR(-59, -2)		-0.138^{*}			-0.060 ($-0.140^{**}(0.034)$		
Distress		0.015 (0			0.052 (0.			-0.024 (0.765)		
Year		-0.090	,		-0.189 (,		-0.018 (0.777)		
Purpose		-0.149 [*]	` '		-0.122 (0.337)		-0.100	5 (0.310)	
Exchange			(0.033)							
Managers		-0.004 (,		0.117 (0.			-0.030(0.766)		
Non-manager affiliated		-0.142	,		-0.043 (,		-0.186(0.158)		
ManagerShare		-0.040 (-0.122 (-0.049(0.695)		
Managers*ManagerShare		-0.071	(0.555)		-0.133 (0.487)		-0.048(0.749)		
N		307			122			185		
Adjusted R ²		0.188			0.137		0.334			

Note: Panels A and B: (Non-) managers represent the private placements involving managers (non-manager investors). Managers with big (small) holdings represent managerial buyers whose initial ownership is above (below) their median ownership. The mean (median) of the differences is tested by t-test (Mann-Whitney U test). Panel C: The dependent variable is the private placement discount measured relative to share price 10 days after the announcement. Fraction is the ratio of shares issued to total shares outstanding after an issue. LnProceeds is the natural logarithm of private placement proceeds. LnMV is the natural logarithm of the stock price on the announcement date multiplied by the number of shares outstanding prior to the placement. BM is the ratio of book value of equity to market value of equity. CAR (-59, -2) is the cumulative abnormal returns over the period beginning 59 days and ending 2 days prior to the private placement announcement. Distress equals to 1 if a firm is under financial distress; otherwise 0. A firm is regarded as under financial distress if it experiences two consecutive years of negative earnings prior to the private placement announcement, or as provided by the Taiwan Economic Journal database, if it faces serious events that threaten its operations. Year equals to 1 if the placement is announced between August 2007 and December 2008, and 0 if the announcement is between January 2001 and July 2007. Purpose equals to 1 if the private placements have a strategic purpose and 0 if the purpose is financial. Exchange equals to 1 if the private equity issuer is listed on TSE and 0 if the issuer is listed on OTC. Managers equals to 1 if the managers of firms participate in private placements; otherwise 0. Non-manager affiliated (managers' relatives, attorneys or the correspondent bank of the firm, current block shareholders of the firm, strategic alliance institutions, and the companies having product market agreements with the firm). ManagerShare equals to 1 if the percentage holding of d

with big managerial ownership. The mean and median shareholding percentages of managers are significantly different between the two categories at the 1% level. The same results are obtained for the TSE and OTC subsamples.

Then we add the variable "ManagerShare" and the interaction term "Managers×ManagerShare" to the regression model²³; the results are presented in Panel C. The coefficient on the interaction term between Managers and ManagerShare in the TSE subsample is negative but not statistically significant. Although the mean

analysis in the previous paragraph leads to the possibility of managerial self-dealing in TSE firms, there is no sufficient evidence supporting the hypothesis in the multivariate results.

5.5. Regression analysis of discount-adjusted abnormal returns

Table 5 reports the cross-sectional regression of discount-adjusted abnormal returns. In all the models of the full sample, the coefficients on *Fraction* are significantly positive, indicating that placing a higher fraction conveys information about the greater magnitude of investment opportunities and future prospects for the issuing firms, which is consistent with Hertzel and Smith's (1993) findings. The significantly positive coefficients on *LnMV*

^{*} Indicates significance at the 10% level.

^{**} Indicates significance at the 5% level.

^{***} Indicates significance at the 1% level.

 $^{^{23}\,}$ ManagerShare is a dummy variable which equals to 1 if the percentage holding of directors and officers before private placements is above the median holding.

Table 5Cross-sectional regression of discount-adjusted abnormal returns.

	Full Sample			TSE			OTC		
	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3	Model 1	Model 2	Model 3
Intercept	-1.573***	-1.624***	-1.588***	-0.673	-0.878^*	-0.602	-2.326***	-2.283***	-2.418***
	(0.000)	(0.000)	(0.000)	(0.112)	(0.099)	(0.153)	(0.000)	(0.000)	(0.000)
Fraction	0.856***	0.815**	0.809**	0.204	0.142	0.082	1.266***	1.236***	1.205***
	(0.008)	(0.015)	(0.018)	(0.776)	(0.856)	(0.917)	(0.000)	(0.000)	(0.000)
LnProceeds	-0.094***	-0.092***	-0.099***	-0.042	-0.038	-0.050	-0.138***	-0.137***	-0.137***
	(0.003)	(0.004)	(0.001)	(0.221)	(0.263)	(0.111)	(0.001)	(0.001)	(0.002)
LnMV	0.153***	0.154***	0.160***	0.067*	0.070*	0.074**	0.226***	0.224***	0.230***
	(0.000)	(0.000)	(0.000)	(0.069)	(0.060)	(0.047)	(0.000)	(0.000)	(0.000)
BM	0.059**	0.055**	0.045*	0.057	0.058	0.031	0.077**	0.072**	0.079*
	(0.014)	(0.024)	(0.069)	(0.109)	(0.105)	(0.322)	(0.033)	(0.043)	(0.068)
CAR (-59, -2)	0.002	-0.000	0.008	0.124	0.112	0.159	-0.042	-0.045	-0.044
,	(0.962)	(1.000)	(0.843)	(0.150)	(0.161)	(0.103)	(0.384)	(0.361)	(0.358)
Distress	0.015	0.010	-0.000	-0.055	-0.072**	-0.096**	0.063	0.057	0.057
	(0.693)	(0.787)	(0.996)	(0.155)	(0.041)	(0.029)	(0.257)	(0.297)	(0.327)
Year	0.072**	0.065**	0.078**	0.022	-0.009	0.029	0.112***	0.107***	0.118***
	(0.022)	(0.035)	(0.018)	(0.727)	(0.858)	(0.650)	(0.003)	(0.004)	(0.003)
Purpose	-0.108**	-0.112***	-0.074^*	-0.095*	-0.086*	-0.068	-0.128**	-0.139**	-0.088
Tarpose	(0.013)	(0.008)	(0.062)	(0.096)	(0.087)	(0.156)	(0.044)	(0.025)	(0.151)
Exchange	-0.052	-0.053	-0.044	(0.000)	(0.007)	(0.150)	(0.011)	(0.020)	(0.101)
Exchange	(0.223)	(0.204)	(0.356)						
Managers	-0.035	-0.026	-0.016	0.017	0.037	0.023	-0.072^{*}	-0.063^{*}	-0.052
Managers	(0.295)	(0.394)	(0.630)	(0.777)	(0.516)	(0.720)	(0.058)	(0.080)	(0.190)
Non-manager affiliated	-0.050	-0.055	-0.035	0.077	0.049	0.057	-0.170	-0.167	-0.147
non manager annacea	(0.527)	(0.489)	(0.644)	(0.175)	(0.437)	(0.284)	(0.219)	(0.226)	(0.261)
Single	-0.026	(0.403)	-0.173***	-0.054	(0.437)	-0.219**	-0.021	(0.220)	-0.174^*
Siligic	(0.369)		(0.006)	(0.186)		(0.014)	(0.560)		(0.071)
Ownership	(0.505)	0.059	(0.000)	(0.100)	0.199	(0.011)	(0.500)	-0.031	(0.071)
Ownership		(0.534)			(0.291)			(0.749)	
∆Ownership		0.185			0.159			0.189	
Zownership		(0.290)			(0.675)			(0.295)	
Single*Fraction		(0.230)	0.553*		(0.073)	0.827		(0.233)	0.565*
Single Traction			(0.059)			(0.191)			(0.061)
Single*BM			0.022			-0.004			0.001)
Siligic bivi			(0.522)			(0.927)			(0.973)
Single*Distress			0.081			0.127			0.083
Single Distress			(0.141)			(0.105)			(0.311)
Single*Exchange			-0.012			(0.103)			(0.511)
Single Exchange			(0.839)						
N	294	294	294	119	119	119	175	175	175
Adjusted R ²	0.133	0.133	0.143	0.041	0.041	0.070	0.245	0.248	0.247

Note: The dependent variable is the discount-adjusted abnormal returns (AR_{adj}) . $AR_{\text{adj}} = [1/(1-\alpha)]$ $[AR] - [\alpha/(1-\alpha)]$ $[(P_0 - P_b)/P_b]$, where α is the ratio of shares placed to shares outstanding after the placement, AR is the abnormal stock return over the event window, P_b is the closing price just one day prior to the event window. Fraction is the ratio of shares issued to total shares outstanding after an issue. LnProceeds is the natural logarithm of private placement proceeds. LnMV is the natural logarithm of the stock price on the announcement date multiplied by the number of shares outstanding prior to the placement. BM is the ratio of book value of equity to market value of equity. CAR(-59, -2) is the cumulative abnormal returns over the period beginning 59 days and ending 2 days prior to the private placement announcement. Distress equals to 1 if a firm is under financial distress; otherwise 0. A firm is regarded as under financial distress if it experiences two consecutive years of negative earnings prior to the private placement announcement, or as provided by the Taiwan Economic Journal database, if it faces serious events that threaten its operations. Year equals to 1 if the placement is announced between August 2007 and December 2008, and 0 if the announcement is between January 2001 and July 2007. Purpose equals to 1 if the private placements have a strategic purpose and 0 if the purpose is financial. Exchange equals to 1 if the private equity issuer is listed on TSE and 0 if the issuer is listed on OTC. $Extit{Managers}$ equals to 1 if the managers of firms participate in private placements; otherwise 0. $Extit{Non-manager}$ $Extit{Mon-manager}$ is listed investors are not managers; otherwise 0. $Extit{Non-manager}$ $Extit{Mon-manager}$ $Extit{Mon-manager}$ $Extit{Mon-manager}$ is listed on OTC. $Extit{Mon-manager}$ $Extit{Mon-manager}$ is listed on TSE and 0 if the affiliated investors are not managers; otherwise 0. $Extit{Mon-manager}$ $Extit{Mon-$

- * Indicate significance at the 10% level.
- ** Indicate significance at the 5% level.
- *** Indicate significance at the 1% level.

show that the greater the announcement abnormal returns are, the larger the firm size. As Wu and Wang (2005) argue, if the asymmetric information about firms is due to their investment opportunities rather than assets-in-place, that is, if the market has uncertainty about the good growth prospects of the firms, the announcement abnormal returns are greater. According to the argument, we conjecture that greater (smaller) announcement effects for large (small) firms reflect the market's concerns about the uncertainty over growth opportunities (assets-in-place). In addition, Malkiel (2003) proposes that in most world markets,

the stocks issued by larger firms generate larger returns because portfolio managers prefer larger firms with more liquidity. 24 Since a small size for private placement firms is common, the investors' concerns about liquidity among these firms may also be a reason for the positive coefficient on LnMV in this study. The significantly

²⁴ Wu (2004) finds that private placement firms are concerned by fewer institutional investors and stock market analysts and have relatively smaller trading volume than public issue firms.

H.-C. Liang, W.-Y. Jang / The Quarterly Review of Economics and Finance xxx (2013) xxx-xxx

positive coefficients on BM, another proxy for growth, indicate that the announcement effects become greater as BM grows larger. This corroborates the view of Wu et al. (2005), who interprets the positive relationship between BM and the announcement effects thus: when the uncertainty of firm value results from the firms' growth

tive relationship between *BM* and the announcement effects thus: when the uncertainty of firm value results from the firms' growth prospects rather than assets in place, the market will respond positively. This also supports the result of significantly positive coefficients on *BM* in private placement discount regression models. If the uncertainty of firm value results from growth prospects, since the growth prospects are more difficult to be valued than assets in place, the investors will demand a greater discount.

The coefficients on *Purpose* are significantly negative. This shows that the market will respond more positively if a firm raises private equity for a financial purpose; as suggested by He (2009), the market will view this improvement in financial condition as an important development for the firm, which would obtain a positive market reaction.

Moreover, the coefficients on *Year* are significantly positive, showing that during the financial crisis period, the private issue announcement implies that firms have the ability to raise funds, which is a sign of market confidence in the firm. Therefore, Besley, Kohers, & Steigner (2007) suggest that a firm having the ability to attract investors to buy private equity shares during unattractive market conditions (e.g., bear markets) delivers a positive signal about its quality.

Models 3 of the full sample and the subsamples report the results after adding the interaction terms. The coefficients on *Single* are significantly negative, and so the single investors who participate in private placements for control benefits should be able to explain the negative market reaction. Besides, the significantly positive interaction between *Single* and *Fraction* shows that the sales to single investors holding a larger fraction of private equity indicate stronger monitoring motives, and so the market responds positively (Models 3 of the full sample and the OTC subsample).

We now discuss the models in the TSE and OTC subsamples. The coefficients on *Fraction, LnProceeds* and *BM* are significant only in the OTC subsample, showing again that information asymmetry variables influence the OTC firms more than the TSE firms.

The coefficient on Distress is significantly negative in Models 2 and 3 of the TSE subsample, showing that when financially distressed TSE firms announce private placements, the market responds negatively. On the other hand, in the OTC subsample, the announcements receive positive responses only during of financial crises (the coefficients on Year are significantly positive). Year and Distress have similar implications, but Distress is based on micro consideration while Year is based on macro consideration. When the capital market is in an unstable condition, announcements by the OTC firms send positive signals to the market, showing that the firms are in sound condition and have growth prospects to attract capital injection from investors, and the market responds positively. When the capital market is in a stable condition, private equity placed by the OTC firms would imply their weak fundraising ability (due to not capable of making public issues), or their unfairness to non-participating shareholders, and so the market responds negatively. However, if the TSE firms face financial distress, implying that the firms are not in sound condition, the market will interpret this quite differently. Additionally, when the managers buy the private placement shares in OTC firms, the market will respond negatively (the coefficients on Managers are significantly negative in Models 1 and 2 of the OTC subsample). This is consistent with Hertzel and Smith's (1993) view that the participations of managers in placements may convey undervaluation of the issuing firms, although the managers may have selfish selfdealing motives. Under the interaction between the two kinds of conflicting incentives of managers, the sales of private placements to managers are viewed a less positive signal than sales to outsiders. Moreover, He (2009) shows that due to the lower visibility of operating condition in OTC firms, the sales to managers may be linked to the managers' pursuit of private benefits. In addition, the managers may be too optimistic about their firms' prospects (Heaton, 2002; Malmendier & Tate, 2005; Wong & Zhang, 2009; Wruck & Wu, 2009), leading to the market's negative reaction to their announcements. All the above arguments provide various explanations for our results.²⁵

Once again, these arguments prove that when the TSE/OTC firms announce their private placements, the market interprets the announcements quite differently. Additionally, the adjusted R^2 is larger in the OTC subsamples for both the discount regression and the discount-adjusted abnormal return regression, showing that when studying the private placements of the TSE and OTC firms, the influential factors are obviously different.

6. Conclusion

This paper studies the price discounts and announcement effects of equity private placements conducted by firms in Taiwan from January 2002 to December 2008, and explores the influential factors. From our empirical results, the information hypothesis and ownership structure hypothesis apply to different conditions. The factors that influence the TSE firms are different from those that influence the OTC firms. Therefore, we consider it more appropriate for future studies to separate the whole sample into subsamples by listed exchanges, such as TSE firms and OTC firms in our study.

In sum, the empirical evidences presented in our study are mixed. The information hypothesis applies to the OTC firms. The discounts serve as compensation for investor's costs of assessing firms, while positive abnormal returns, regarded as the OTC firms' message conveying good news through private placements, imply asymmetric information from uncertainty about growth prospects. On the other hand, the empirical results show that some of our findings support an information explanation and some support a monitoring explanation in the case of the TSE firms. It seems that there are different motives behind the exchange-listed firms placing equity privately. The effects induced by the different motives might partially offset each other.

As suggested by Hertzel and Smith (1993), changes in firm value surrounding private placements may arise from the market's evaluation of firm assets and investment opportunities; when firms face limited financial slack, they may raise funds through private placements. Therefore, the importance of the information asymmetry hypothesis in our results may partially result from the fact that a rather large number of sample firms have faced financial distress (accounting for 48% of all sample firms, Panel B of Table 1).

Furthermore, single investors participate in private placements in order to obtain control premium and pursue control benefits. The evidence substantiating this claim is fairly conclusive in TSE firms, where investors are willing to purchase private equity shares at higher prices. In contrast to many limitations of public issues, the private issues by firms have effectively more flexibility for financing. However, it is also likely that the capital market provides a convenient method for managers to pursue self-interest. Besides, the trend for firms with net loss to conduct private placements following capital reduction is upward in Taiwan, and private placement shares are often offered at a discount. All of these

Please cite this article in press as: Liang, H. -C., & Jang, W.-Y. Information asymmetry and monitoring in equity private placements. *The Quarterly Review of Economics and Finance* (2013), http://dx.doi.org/10.1016/j.qref.2013.05.009

 $^{^{25}}$ We also substitute Affiliated for Managers and Non-manager affiliated to conduct regression analysis. Because the results are the same, and Managers and Non-manager affiliated provide more detailed information, we do not show the regression containing Affiliated.

factors are inevitably associated with the firms selling their equity at unfavourable prices to benefit specific investors. In order to prevent these negative side effects of private placements, the Financial Supervisory Commission seeks to enhance the supervision of the private placement system by amending and promulgating the Notice for Private Placement of Marketable Securities by Publicly Offered Companies in 2010, which stipulates restrictions on private placements by profitable companies, and the setting of reference prices²⁶ (Articles 3 and 4). Thus, not all the firms are allowed to place equity privately. This implies that private placements in Taiwan are the major financing method for financially distressed firms, and could be the last resort for firms to survive. We believe that the ongoing amended regulations, matched with a practical environment, will reflect the firm value properly through the market mechanisms and make private placements intrinsically more meaningful. After all, we expect private placements to facilitate the financing activity of firms, and not the pursuit of self-interest.

References

- Alli, K. L., & Thompson, D. J., II. (1993). The wealth effects of private stock placements under regulation D. *Financial Review*, 28, 329–350.
- Anderson, H. D., Rose, L. C., & Cahan, S. F. (2006). Differential shareholder wealth and volume effects surrounding private equity placements in New Zealand. *Pacific-Basin Finance Journal*, 14, 367–394.
- Antweiler, W., & Frank, M. Z. (2006). Do US stock markets typically overreact to corporate news stories? Available at SSRN: http://ssrn.com/abstract=878091 (accessed August 2006)
- Barclay, M. J., & Holderness, C. G. (1989). Private benefits from control of public corporations. *Journal of Financial Economics*, 25, 371–395.
- Barclay, M. J., Holderness, C. G., & Sheehan, D. P. (2007). Private placements and managerial entrenchment. *Journal of Corporate Finance*, 13, 461–484.
- Barclay, M. J., & Smith, C. W., Jr. (1995). The maturity structure of corporate debt. *Journal of Finance*, 50, 609–631.
- Besley, S., Kohers, N., & Steigner, T. (2007). Private placements of common equity and the industry rival response. Applied Financial Economics, 17, 559–568.
- Butler, A. W., Grullon, G., & Weston, J. P. (2005). Stock market liquidity and the cost of issuing equity. *Journal of Financial & Quantitative Analysis*, 40, 331–348.
- Carpentier, C., L'Her, J.-F., & Suret, J.-M. (2008). Does securities regulation constrain small business finance? An empirical analysis. *Small Business Economics*, 31, 363–377
- Chen, S.-S., Ho, K. W., Lee, C.-f., & Yeo, G. H. H. (2002a). Wealth effects of private equity placements: evidence from Singapore. *Financial Review*, 37, 165–184.
- Chen, S.-S., Ho, K. W., Lee, C.-F., & Yeo, G. H. H. (2002b). Long-run stock performance of equity-issuing firms: the case of private placements in Singapore. *Review of Pacific Basin Financial Markets & Policies*, 5, 417–438.
- Chou, D.-W., Gombola, M., & Liu, F.-Y. (2009). Long-run underperformance following private equity placements: the role of growth opportunities. *Quarterly Review of Economics and Finance*, 49, 1113–1128.
- Claessens, S., Djankov, S., & Lang, L. H. P. (2000). The separation of ownership and control in East Asian corporations. *Journal of Financial Economics*, 58, 81–112.
- Coffee, J. C. J. (1995). Re-engineering corporate disclosure: The coming debate over company registration. Washington and Lee Law Review, 52, 1143–1188.
- Corwin, S. A. (2003). The determinants of underpricing for seasoned equity offers. Journal of Finance, 58, 2249–2279.
- Cronqvist, H., & Nilsson, M. (2005). The choice between rights offerings and private equity placements. *Journal of Financial Economics*, 78, 375–407.
- Marciukaityte, D., Szewczyk, S. H., & Varma, R. (2005). Investor overoptimism and private equity placements. *Journal of Financial Research*, 28, 591–608.
- Dann, L. Y., & DeAngelo, H. (1988). Corporate financial policy and corporate control: A study of defensive adjustments in asset and ownership structure. *Journal of Financial Economics*, 20, 87–127.
- Demsetz, H., & Lehn, K. (1985). The structure of corporate ownership: causes and consequences. *Journal of Political Economy*, 93, 1155–1177.
- Fenn, G. W. (2000). Speed of issuance and the adequacy of disclosure in the 144A high-yield debt market. *Journal of Financial Economics*, 56, 383–405.
- Folta, T. B., & Janney, J. J. (2004). Strategic benefits to firms issuing private equity placements. Strategic Management Journal, 25, 223–242.
- Greenspan, A. (2010). *The crisis*. http://www.brookings.edu/~/media/Files/ Programs/ES/BPEA/2010_spring.bpea_papers/spring2010_greenspan.pdf
- He, G. (2009). The contents of news and the offering price of the equity private placement. (Master's Thesis).
- ²⁶ The competent authority requires that a firm must not use the private placement method of issuing securities to raise capital when the firm has a net profit and no accumulated deficit for the most recent fiscal year.

- Heaton, J. B. (2002). Managerial optimism and corporate finance. Financial Management, 31, 33–45.
- Hertzel, M., Lemmon, M., Linck, J. S., & Rees, L. (2002). Long-run performance following private placements of equity. *Journal of Finance*, 57, 2595–2617.
- Hertzel, M., & Rees, L. (1998). Earnings and risk changes around private placements of equity. *Journal of Accounting Auditing & Finance*, 13, 21–35.
- Hertzel, M., & Smith, R. L. (1993). Market discounts and shareholder gains for placing equity privately. *Journal of Finance*, 48, 459–485.
- Hung, C.-W., Shiu, C.-Y., & Wei, H.-S. (2012). Stock liquidity and the choice between public offerings and private placements. *Journal of Financial Studies*, 20, 1–22.
- Hsieh, W.-T., & Wu, T. C. (2010). Determinants and market reactions: public offering vs. private placement. *International Journal of Accounting Studies*, 51, 73–106
- Jang, W.-Y., Tsai, M.-S., & Kuo, Y.-F. (2010). The announcement effects of private equity placement and ownership structure changes evidence from listed companies in Taiwan. *Journal of Financial Studies*, *18*, 91–136.
- Jensen, M. C. (1986). Agency costs of free cash flow, corporate finance, and takeovers. American Economic Review, 76, 323–329.
- Jensen, M. C., & Meckling, W. H. (1976). Theory of the firm: managerial behavior, agency costs and ownership structure. *Journal of Financial Economics*, 3, 305–360.
- Klein, B., Crawford, R. G., & Alchian, A. A. (1978). Vertical integration, appropriable rents, and the competitive contracting process. *Journal of Law and Economics*, 21, 297–326.
- Krishnamurthy, S., Spindt, P., Subramaniam, V., & Woidtke, T. (2005). Does investor identity matter in equity issues? Evidence from private placements. *Journal of Financial Intermediation*, 14, 210–238.
- Krishnaswami, S., Spindt, P. A., & Subramaniam, V. (1999). Information asymmetry, monitoring, and the placement structure of corporate debt. *Journal of Financial Economics*, 51, 407–434.
- La Porta, R., Lopez-de-Silanes, F., & Shleifer, A. (1999). Corporate ownership around the world. *Journal of Finance*, 54, 471–517.
- Lee, C.-F., & Wu, Y. (2009). Two-stage models for the analysis of information content of equity-selling mechanisms choices. *Journal of Business Research*, 62, 123–133
- Leland, H. E., & Pyle, D. H. (1977). Informational ssymmetries, financial structure, and financial intermediation. *Journal of Finance*, 32, 371–387.
- LeRoy, D. B., & Graham, J. E. (2005). Equity private placements, liquid assets, and firm value. Journal of Economics and Finance, 29, 321–336.
- Lin, C.-J., & Chang, C.-C. (2009). Abnormal change of board members, family firms and fraud. *International Journal of Accounting Studies*, 48, 11–33.
- Lin, J.-C., & Howe, J. S. (1990). Insider trading in the OTC market. *Journal of Finance*, 45, 1273–1284.
- Malkiel, B. G. (2003). The efficient market hypothesis and its critics. *Journal of Economic Perspectives*, 17, 59–82.
- Malmendier, U., & Tate, G. (2005). CEO overconfidence and corporate investment. *Journal of Finance*, 60, 2661–2700.
- Mathew, P. G. (2002). Long-horizon seasoned equity offerings performance in Pacific rim markets. *Review of Financial Economics*, 11, 317–333.
- Maynes, E., & Pandes, J. A. (2011). The wealth effects of reducing private placement resale restrictions. *European Financial Management*, 17, 500–531.
- Miller, M. H., & Rock, K. (1985). Dividend policy under asymmetric information. Journal of Finance, 40, 1031–1051.
- Morck, R., Shleifer, A., & Vishny, R. W. (1988). Management ownership and market valuation: An empirical analysis. *Journal of Financial Economics*, 20, 293-315.
- Myers, S. C., & Majluf, N. S. (1984). Corporate financing and investment decisions when firms have information that investors fo not have. *Journal of Financial Economics*, 13, 187–221.
- Scholes, M., & Williams, J. (1977). Estimating betas from nonsynchronous data. *Journal of Financial Economics*, 5, 309–327.
- Sharpe, İ. G., & Woo, L.-A. E. (2005). Corporate control, expected underpricing, and the choice of issuance mechanism in unseasoned equity markets. *Journal of Corporate Finance*, 11, 716–735.
- Shleifer, A., & Vishny, R. W. (1986). Large shareholders and corporate control. *Journal of Political Economy*, 94, 461–488.
- Shleifer, A., & Vishny, R. W. (1997). A survey of corporate governance. *Journal of Finance*, 52, 737–783.
- Silber, W. L. (1991). Discounts on restricted stock: The impact of illiquidity on stock prices. *Financial Analysts Journal*, 47, 60–64.
- Spiess, D. K., & Affleck-Graves, J. (1995). Underperformance in long-run stock returns following seasoned equity offerings. *Journal of Financial Economics*, 38, 243–267.
- Stulz, R. (1990). Managerial discretion and optimal financing policies. *Journal of Financial Economics*, 26, 3–27.
- Tan, R. S. K., Chng, P. L., & Tong, Y. H. (2002). Private placements and rights issues in Singapore. Pacific-Basin Finance Journal, 10, 29-54.
- Wayne, H. M., & Hailu, R. (1991). Premiums paid in block transactions. Managerial and Decision Economics (1986–1998), 12, 511–517.
- Wong, M. H. F., & Zhang, F. (2009). CEO optimism, analyst rationality, and earnings gorecast bias. Available at SSRN: http://ssrn.com/abstract=814085 (accessed 05.12.09)

H.-C. Liang, W.-Y. Jang / The Quarterly Review of Economics and Finance xxx (2013) xxx-xxx

- Wruck, K. H. (1989). Equity ownership concentration and firm value: Evidence from private equity financings. *Journal of Financial Economics*, 23, 3–28.
- Wruck, K. H., & Wu, Y. (2009). Relationships, corporate governance, and performance: evidence from private placements of common stock. *Journal of Corporate Finance*, 15, 30–47.
- Wu, X., & Wang, Z. (2005). Equity financing in a Myers–Majluf framework with private benefits of control. *Journal of Corporate Finance*, 11, 915–945.
- Wu, X., Wang, Z., & Yao, J. (2005). Understanding the positive announcement effects of private equity placements: new insights from Hong Kong data. *Review of Finance*, 9, 385–414.
- Wu, Y. (2004). The choice of equity-selling mechanisms. *Journal of Financial Economics*, 74, 93–119.
- Yeh, Y.-H., Lee, T.-S., & Woidtke, T. (2001). Family control and corporate governance: Evidence from Taiwan. *International Review of Finance*, 2, 21–48.

16



این مقاله، از سری مقالات ترجمه شده رایگان سایت ترجمه فا میباشد که با فرمت PDF در اختیار شما عزیزان قرار گرفته است. در صورت تمایل میتوانید با کلیک بر روی دکمه های زیر از سار مقالات نیز استفاده نمایید:

🗸 لىست مقالات ترجمه شده

🗸 لیست مقالات ترجمه شده ر ایگان

✓ لیست جدیدترین مقالات انگلیسی ISI

سایت ترجمه فا ؛ مرجع جدیدترین مقالات ترجمه شده از نشریات معتبر خارجی